Author's response to reviews

Title: Experience of Nurses with using eHealth in GilgitBaltistan, Pakistan: a qualitative study in primary and secondary healthcare.

Authors:

Saleema Gulzar (saleema.gulzar@aku.edu)
Shariq Khoja (skhoja@mhealthalliance.org)
Afroz Sajwani (afroz.sajwani@gmail.com)

Version: 7 Date: 2 January 2013

Author's response to reviews: see over
Jan 02, 2013

BMC Nursing
BioMed Central
236 Gray's Inn Road
London WC1X 8HB
United Kingdom

Subject: Revision of manuscript entitled ‘Experience of Nurses with using eHealth in Gilgit Baltistan, Pakistan: a qualitative study in primary and secondary healthcare’

Dear Tom,

We have revised the manuscript as per reviewers’ comments. The point-by-point answers to the concerns are mentioned below:

- Overall, the manuscript has been revised for language.
- The suggested wording on the ethics approval has been corrected.
- Methods section is revised clearly stating the type of study and how participants were selected representing workforce in the district.
- Description of ‘latent analysis’ has been revised.
- The limitations of this study are addressed in the last paragraph of the Discussion section on page 13.
- The finding that introducing ehealth can necessitate a change in the organization of health services has been addressed in the discussion section in the last paragraph on page 12.

We look forward to the next round of review, and your positive feedback.

Sincerely,

Shariq Khoja MBBS. MSc. PhD (e-Health)
Director- AKDN eHealth Resource Centre
The Aga Khan University
3rd Parklands Avenue
Nairobi, PO Box 30270-00100
Kenya
Phone: 254 20 3662236
Fax: 254 20 3744956
Email: skhoja@mhealthalliance.org